Author's response to reviews

Title: Long-term prognosis of clinically early IgA nephropathy is not always favorable

Authors:

Hajeong Lee (mdhijlee@gmail.com)
Jin Ho Hwang (dennyjinho@gmail.com)
Jin Ho Paik (paikihmd@gmail.com)
Hyun Jin Ryu (hdutopia@naver.com)
Dong Ki Kim (dkkim73@gmail.com)
Ho Jun Chin (mednep@snubh.org)
Yun Kyu Oh (yoonkyuoh@gmail.com)
Kwon Wook Joo (junephro@paran.com)
Chun Soo Lim (cslimjy@snu.ac.kr)
Yon Su Kim (yonsukim@snu.ac.kr)
Jung Pyo Lee (nephrolee@gmail.com)

Version: 3 Date: 9 April 2014

Author's response to reviews: see over
Response Letter
Prof Guy Neild, Editor-in-Chief
BMC Nephrology

Dear Editor,

Thank you for giving us the opportunity to revise our manuscript entitled “Long-term prognosis of clinically early IgA nephropathy is not always favorable” by H Lee et al. (manuscript ID: MS: 1655192384118013) for consideration for publication in BMC Nephrology.

The reviewer’s comments were very helpful and we have revised our manuscript accordingly. We have addressed the reviewer’s specific points below. We believe that the revised manuscript has been significantly improved as a result of the reviewer’s comments, and we hope that it will now be acceptable for publication in BMC Nephrology.

Yours sincerely,

Jung Pyo Lee, MD, PhD
Assistant Professor
Department of Internal Medicine
Seoul National University Boramae Medical Center
425 Sindaeubang 2-dong
Dongjak-gu, Seoul 156-707, Republic of Korea
Tel. +82-2-870-3206, Fax +82-2-870-2826
E-mail: nephrolee@gmail.com